Author’s response to reviews

Title: Coreopsis tinctoria Nutt ameliorates high glucose-induced renal fibrosis and inflammation via the TGF-β1/SMADs/AMPK/NF-κB pathways

Authors:
Lan Yao (ylyl1026@sina.com)
Jie Li (alexli91@yahoo.com)
Lin lin Li (442338720@qq.com)
Xin xia Li (lxx6668@163.com)
Rui Zhang (603210439@qq.com)
Yu jie Zhang (835580053@qq.com)
Xin Min Mao (mxm3277@139.com)

Version: 2 Date: 15 May 2018

Author’s response to reviews:

A point-by-point response letter

1. The English language still needs some corrections, for instance see the line No, 132
   English language was corrected line 132,line133 page6

2. Line 108, Remove the bracket
   The bracket was removed line 108 page5

3. The word "in vitro" should be as per standard format
   All "in vitro""in vivo" has been as per standard format line 45,47,83,111,146,284,288,327,328 etc.

4. Line 273, the plant name should be Rumex hastatus in citation 33 and secondly add a sentence for citation 34 also.
   the plant name of Rumex hastatus has been changed and a sentence for citation 34 was added line 272,273,274
5. The conclusion section still need to be refined
   Line 386,387,388

6. In the figure 7 the size of asterisks and hash should be reduced
   Yes size of asterisks and hash has been reduced in the figure

7. The plants' names in the references should also be italicized for instance see reference 20, 21, 22 etc
   plants' names in the references 20, 21, 22,34,35 have been italicized

"The authors' response letter has been included as a supplementary file"